Author's response to reviews

Title: Comparing the frequency of common genetic variants and haplotypes between carriers and non-carriers of BRCA1 and BRCA2 deleterious mutations in Australian women diagnosed with breast cancer before 40 years of age

Authors:

Lidija Turkovic (lidija78@yahoo.com)
Lyle C Gurrin (lgurrin@unimelb.edu.au)
Melanie Bahlo (bahlo@wehi.EDU.AU)
Gillian S Dite (g.dite@unimelb.edu.au)
Melissa C Southey (msouthey@unimelb.edu.au)
John L Hopper (j.hopper@unimelb.edu.au)

Version: 5 Date: 12 August 2010

Author's response to reviews:

Dear Editor,

I have made the formatting changes to the manuscript as requested.

Regards,

Lidija Turkovic